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10 July 2011
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 27 of 2011, which covers the period from 03 through 09 July 2011.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 377.7 to 433.2 kcfs.  Daily average project tailwater ranged from 26.9’ to 29.1’ msl.  Daily average powerhouse forebay elevation ranged from 73.3’ to 74.2’ msl.  Daily average river temperature ranged from 62 to 63ºF. Secchi disk measurements ranged from 2.5’ to 4.0’.
1.2 Spill Bays 1 and 18, which provide attraction flow for Cascades Island and B-branch, respectively, are open 6” during daylight hours.  Daily average spill ranged from 161.9 to 218.6 kcfs.
1.3 Unit Operation.  Powerhouse One remains the priority powerhouse.
1.3.1 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.

1.3.2 Unit 15 went out of service at 0717 on 15 June for four year overhaul.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.  This week required fish units to be placed on standby to float debris during daylight hours due to excessive drawdown.  
1.5 Gatewell drawdowns are measured on Wednesdays and more frequently as needed.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  No AWS valves were closed for cleaning this reporting week.
2.2 STS/VBS inspections.  No STS/VBS inspections occurred this reporting week. STSs remain dogged off.
Table 1.  STS and VBS inspections.
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	U11 remains out of service.

	12-18
	NA
	NA
	Inspection scheduled for week of 25 July.


2.3 Dewatering and Fish Salvage.  No fish salvage activities occurred this reporting week.
3 RESEARCH  
3.1 Battelle - survival study.  Monitoring University of Idaho acoustic tagged adult lamprey.   
3.2 CRITFC – AFF.  Annual chinook, coho, sockeye and steelhead sampling in the AFF continues.
3.3 CRITFC - Sea lion.  Acoustic receivers for sea lion tags remain anchored in the BRZ.  
3.4 NOAA/PSMFC - lamprey.  LPS count network improvements moving forward.  Traps deployed at PH2 monoliths in tailrace and AFF. 
3.5 NOAA – Pathology Study. A total of 28 subyearling chinook were collected for NOAA pathology studies this week.
3.6 PSMFC – Smolt Monitoring Program.  Juvenile fish condition and index monitoring continues.  Fallbacks recorded crossing the separator bars:  9 steelhead, 28 salmon, and 38 unknown salmonids.  Gas Bubble Trauma (GBT) examinations occur two days per week at the SMF during spill.  Please go to http://www.fpc.org/CurrentDaily/Gbtsum.txt for further details.  
3.7 University of Idaho.  Tagging lamprey in the AFF continues.

3.8 USDA - Avian hazing. Hazing continues.

3.9 WDFW - Fish Counting.  Visual counting continues.
4 FISHWAYS 
4.1 Project biologists inspected 03 through 09 July. NOAA inspected 08 July.
4.2 Fish Passage Plan Observations.
5 Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	7/3-7/6
	Spillway
	Spill gate #13 position
	In local for overload problems
	Ops adjusted
	

	BRADFORD ISLAND (BI) – Nothing to report.

	CASCADES ISLAND (CI) – Nothing to report.

	WASHINGTON SHORE (WS)

	7/3, 7/5
	PH2 entrances
	Submergence <13’
	Necessary to maintain differential
	
	

	7/9
	DSM2
	Mechanical screen cleaner OOS
	Broken chain
	TR and adjusted air burst system to heavy debris
	11-196891


5.1.1 PH1 PCC diffuser FG2-19 is partially open.  Repairs scheduled for 2011/2012 in water work period.  

5.1.2 A new mechanical velocity meter has been fabricated for PH2 and is scheduled for installation.
5.1.3 The SA-24 board that controls the Powerhouse Two adult ladder components is in need of replacement. Work has begun on replacing the PLC to allow for fully automated operation of the PH2 adult fishway.
5.2 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.

Table 3.  Bonneville Fishway Temperatures

	Date
	WA. Shore average
	BI

average

	7/3
	60.6
	61.0

	7/4
	60.6
	61.0

	7/5
	60.8
	61.2

	7/6
	61.2
	61.7

	7/7
	61.5
	62.0

	7/8
	61.5
	61.5

	7/9
	61.6
	62.0


5.3 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
5.4 Adult Fishways. 
5.4.1 Calibration was checked on 08 July.
5.4.2 10 of 12 floating orifice gates (FOGs) are installed with sea lion barriers in place.  Two gates remain out for maintenance. 
5.4.3 SLEDs were removed from all entrance gates as of 05 July. 
5.4.4 The AFF remains watered up.
5.4.5 Cascades Is. and Bradford Is. ladders were put into operation for shad passage on 14 June. 
5.4.6 One section of diffuser grating between weir 34 and 35 came loose during high tailwater and was removed on 14 June.  Please see 16 June MOC for more information.
5.4.7 The Bradford Is. AWS was dewatered and the ladder had reduced flow to remove 100 - 200 sockeye trapped in the AWS section downstream of the picket leads on 30 June.    Please see 30 June Bonneville Fisheries MFR for more information. 
5.5 Juvenile Fishways.
5.5.1 DSM2 remains in service. 
5.5.2 All STSs are currently pulled.
5.5.3 The PH1 ice and trash sluiceway (ITS) remains in service.  Manual chaingates 1A and 1B remain open and are both set to 70’ msl. Automated chaingates 3B, 6C and 10B are open and functioning in auto.
5.5.4 The hydrocannon was returned to service on 06 July.  
5.5.5 The B2CC remains in service.  Five avian lines are broken due to high tailwater and three remain.  
5.5.6 Operators manually adjusted spill gates 13 and 14 to flush trash on 23 June.  Both bays encountered problems and were dogged at the most appropriate position for the spill pattern. Bay 14 had problems with the brakes and was repaired on 02 July.  Bay 13 had problems with the overloads and was dogged at 4.9 ft on 05 July. It was repaired and returned to service 06 July.  Bay 17 was moved from the dogged position of 2.9 ft to 4.9 ft on 28 June to get closer to FPP spill pattern positions.     
5.6 Lamprey Fishways.
5.6.1 The Cascades Is. entrance weir remains installed.  The CI LPS is in operation on weekdays.
5.6.2  The NDE ramp remains in the lowered position and is not currently operating.
5.6.3 WA Shore and Bradford Island LPSs remain watered up.
6 CONSTRUCTION
6.1 PH1 Gantry 4.  New crane in August, 2011.
6.2 B2CC Hoist. Construction scheduled to begin fall of 2011.
7 HAZMAT, SPILLS AND CLEANUP. Nothing to report.
8 TELETYPES AND OTHER NOTEWORTHY ITEMS
8.1 A teletype sent on 19 May requests that BON operate Powerhouse One as the priority powerhouse. Continue this operation until further notice. The goal of this operation is to minimize debris loading that is contributing to increased descaling and mortality of juvenile fish at Powerhouse Two. Additionally, operate units 14, 15, and 16 without Submersible Traveling Screens. Vertical Barrier Screens (VBSs) will remain installed for these units until debris loading causes a drawdown of 1.5 ft. Once VBSs satisfy this criterion, remove VBSs in units 14, 15, and 16. When selecting operating range within the 1% range of the PH2 screened units (12, 13, 17, and 18) make efforts to the extent practicable to achieve the following two objectives: 1) Do not exceed the 1.5 ft head differential prior to the next cleaning and 2) Maximize discharge at each unit. Objective 1 takes priority over objective 2.
8.2 A teletype sent on 15 June requests summer spill for juvenile fish passage with two treatments of 85 kcfs during the day and 121 kcfs at night or 95 kcfs for 24 hours.  Effective 16 June at 0430 hours and continuing until 31 August at 2400 hours, spill in accordance with the 2011 Fish Passage Plan and the 2011 Summer Fish Operations Plan, Appendix E of the Fish Passage Plan.  
8.3 A teletype sent on 01 July requests that Bonneville pool be operated as follows:  Operate within a 1.5 foot band, soft constraint, from 0600 hours Tuesday, 05 July until 1800 hours Friday, 08 July.  The goal of this operation is to limit pool fluctuation during treaty fishing.
8.4 A teletype sent on 01 July requests that Bonneville pool be operated as follows:  Operate within a 1.5 foot band, soft constraint, from 0600 hours Monday, 11 July until 1800 hours Friday, 15 July.  The goal of this operation is to limit pool fluctuation during treaty fishing.
8.5 A teletype sent on 01 July requests that BON operate, as soon as conditions allow, so that hourly average releases are at or below 460 kcfs. 
8.6 A teletype sent on 07 July cancels teletype 8.5, since operations for system flood control are no longer needed.

8.7 A teletype sent on 07 July request that BON forebay be maintained between elevation 73.0’ and 74.0’ as a soft constraint from 1000 to 1500 hrs each day Wed-Fri, 13-15 July. This operation is needed for a kite board event at the port of Hood River.
8.8 A teletype sent on 07 July requests that BON forebay be maintained between elevation 75.5’ and 76.5’ as a hard constraint and 76.0’ and 76.5’ as a soft constraint from 1400 to 2000 hrs on Saturday, 06 August for a sail boat race. This request allows sail boats to be pulled from the water at the Dalles boat ramp after racing.

8.9 A teletype sent on 08 July requests that BON forebay be maintained between elevation 74.0’ and 75.5’ as a soft constraint, Saturday 16 July and Sunday 17 July, from 1000 to 1700 hrs. This operation will assure unobstructed access for large canoes to launching and landing sites during the Gore Outrigger races near Stevenson, WA.
JERRY CARROLL
Operations Project Manager
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